Author’s response to reviews

Title: A novel technique for repositioning a nasobiliary catheter from the mouth to nostril in Endoscopic Retrograde Cholangiopancreatography

Authors:

Seong Ji Choi (coolandy03@hanmail.net)
Jae Min Lee (jmlee1202@gmail.com)
Hyuk Soon Choi (mdkorea@gmail.com)
Eun Sun Kim (silverkes@naver.com)
Bora Keum (borakeum@hanmail.net)
Yeon Seok Seo (gandorie@gmail.com)
Yoon Tae Jeen (ytjeen@korea.ac.kr)
Hong Sik Lee (hslee60@korea.ac.kr)
Hoon Jai Chun (drchunhj@gmail.com)
Soon Ho Um (umsh@korea.ac.kr)
Chang Duck Kim (kumcge@korea.ac.kr)
Chi Hyuk Oh (coolsoulmate@naver.com)

Version: 3 Date: 11 Dec 2019

Author’s response to reviews:

Dear Jashan Abraham,

We are delighted to be potentially acceptable for publication in BMC Gastroenterology.
As you have required, we made revisions.

Editor Comments:

1. Consent for publication
We obtained the consent for publication from the patient in the supplementary video using BMC patient consent form in Korean language. We also added the comment in the “consent for publication” section as follows: Written informed consent was obtained from the patient for the
publication of supplementary video. A copy of the written consent is available for review by the editors of this journal.
If you need a copy of the consent obtained, please let us know.

2. Role of Funders
As you have requested, we added the following sentence in the “Funding” section: The funding body had no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

3. Formatting
We are sorry that we did not keep the format. We renamed Introduction to Background as you requested.

4. Clean manuscript
We have cleared out any irrelevant changes for the publication including highlights in both manuscript and table.

Thank you again for your effort to improve our article.
If you have any request or questions, or if there’s anything we missed, please don’t hesitate to contact us.

Sincerely,
Jae Min Lee